Author's response to reviews

Title: A novel naproxen derivative capable of displaying anti-cancer and anti-metastatic properties against human breast cancer cells

Authors:

Jolly Deb (bcjd@iacs.res.in)
Joydeb Majumder (ocjm@iacs.res.in)
Siddhartha Sankar Jana (bcssj@iacs.res.in)

Version: 3
Date: 20 February 2014

Author's response to reviews:

1. Email addresses of all co-authors have been added in the title page
2. Tables have been placed at the end of the manuscript after the references section.